Author's response to reviews

Title: MicroRNA-100 is a potential molecular marker of non-small cell lung cancer and functions as a tumor suppressor by targeting polo-like kinase 1

Authors:

Jing Liu Dr (liujing_nanjing@163.com)
Kai-Hua Lu Dr (lk_h_nj@163.com)
Zhi-Li Liu Dr (zl_l_nanjing@yeah.net)
Ming Sun Dr (sunming_nanjing@yahoo.com.cn)
Wei De Dr (dewe_3naji@yahoo.com.cn)
Zhao-Xia Wang Dr (wangzhaox@yahoo.com.cn)

Version: 4 Date: 6 November 2012

Author's response to reviews: see over
Dear Editors:

Thank you for reviewing our manuscript and thank you very much for your comments and suggestion on this work (MicroRNA-100 is a potential molecular marker of non-small cell lung cancer and functions as a tumor suppressor by targeting polo-like kinase 1).

We have revised our manuscript according to editors’ and reviewers’ comments as follow:


If there are some other problems about this manuscript, contact us in time. Thanks a lot.

Best your regards

Sincerely yours

Zhao-Xia Wang

Jing Liu†, Kai-Hua Lu2,3†, Zhi-Li Liu1, Ming Sun4, Wei De4, Zhao-Xia Wang1*

*Correspondence: wangzhaox@yahoo.com.cn.

†Contributed equally.

1Department of Oncology, The Second Affiliated Hospital of Nanjing Medical University, 121 Jiangjiayuan Road, Nanjing 210011, Jiangsu, P.R. China.

2Immunology and Reproductive Biology Lab of Medical School and State Key Laboratory of Pharmaceutical Biotechnology, Nanjing University, Nanjing 210093, Jiangsu, P.R. China.

3Department of Oncology, The First Affiliated Hospital of Nanjing Medical University, Nanjing 210029, Jiangsu, P.R. China.

4Department of Biochemistry and Molecular Biology, Nanjing Medical University, Nanjing 210029, Jiangsu, P.R. China.